Author’s response to reviews

Title: Challenges in a six-phase process of questionnaire adaptation: findings from the French translation of the Integrated Palliative care Outcome Scale

Authors:

Anca-Cristina Sterie (anca-cristina.sterie@chuv.ch)
Mathieu Bernard (mathieu.bernard@chuv.ch)

Version: 2 Date: 02 Apr 2019

Author’s response to reviews:

Editor Comments:

1. Thank you for stating that consent was obtained orally. However, we ask that in the Ethics approval section of the Declarations, please include reasoning for this and state if the ethics committee approved this method of obtaining consent.

Our response

We have updated the Ethics section by specifying that the ethics committee approved of using only oral consent for participation to the transcultural adaptation and the rationale of this decision.